Author's response to reviews

Title: Prognostic significance of multiple kallikreins in high-grade astrocytoma

Authors:

Kristen L Drucker (drucker.kristen@mayo.edu)
Caterina Gianinni (giannini.caterina@mayo.edu)
Paul A Decker (decker.paul@mayo.edu)
Eleftherios P Diamandis (ediamandis@mtsinaion.ca)
Isobel A Scarisbrick (scarisbrick.isobel@mayo.edu)

Version: 2
Date: 13 January 2015

Author's response to reviews: see over
Dr. Dafne Solera  
Executive Editor  
BMC Cancer  

January 13, 2015  

Dear Dr. Solera,  

The manuscript “The Prognostic Significance of Multiple Kallikreins in High Grade Astrocytoma” (MS: 2074495428156407) has been updated according to the requests of the Journal Editorial office.  

1. The section for “Competing Interests” has been added to the manuscript. (line 314-315)  
2. A sentence regarding the use of prior written consent was also added to the Methods section. (line 111)  

Thank you very much for your timely response to our submission.  

Sincerely,  

Isobel A. Scarisbrick Ph.D.  
Associate Professor of Physiology  
Neurobiology of Disease Program  
Department of Physical Medicine & Rehabilitation  
642B Guggenheim  
Mayo Clinic and Foundation  
200 First St. SW, Rochester MN, 55905  
Work Tel: 507-284-0124 (538-3348)